Author's response to reviews

Title: Identification of gene targets against dormant phase Mycobacterium tuberculosis infections

Authors:

Dennis J Murphy (Dennis.6.Murphy@gsk.com)
James R Brown (James.R.Brown@gsk.com)

Version: 6 Date: 24 July 2007

Author's response to reviews:

Please find attached a revised manuscript with the reference numbers hopefully resolved. The problem is with our version of Reference Manager which seems to sometime make duplicate reference entries. The total number of references should be 120.

Thank you

Jim Brown